Author’s response to reviews

Title: REALizing and improving management of stable COPD in China: a multi-center, prospective, observational study to realize the current situation of COPD patients in China (REAL) - rationale, study design, and protocol

Authors:

Ting Yang (dryangting@qq.com)
Baiqiang Cai (caibq2009@hotmail.com)
Bin Cao (caobin_ben@163.com)
Jian Kang (kangjian58@163.com)
Fuqiang Wen (wenfuqiang@126.com)
Wanzhen Yao (yaowanzhen@126.com)
Jinping Zheng (jpzhenggy@163.com)
Xia Ling (xia.ling@astrazeneca.com)
Hongyan Shang (Shirley.shang@astrazeneca.com)
Chen Wang (wangchen66366@163.com)

Version: 2 Date: 29 Nov 2018

Author’s response to reviews:

Ethics:

If your study involves humans, human data or animals, then your article should contain an ethics statement which includes the name of the committee that approved your study.

If ethics was not required for your study, then this should be clearly stated and a rationale provided.

- This information is provided on page 14
Consent:

If your article is a prospective study involving human participants then your article should include a statement detailing consent for participation.

If individual clinical data is presented in your article, then you must clarify whether consent for publication of these data was obtained.

- This information is provided on page 14

Availability of supporting data:

BioMed Central strongly encourages all data sets on which the conclusions of the paper rely be either deposited in publicly available repositories (where available and appropriate) or presented in the main papers or additional supporting files, in machine-readable format whenever possible. Authors must include an Availability of Data and Materials section in their article detailing where the data supporting their findings can be found. The Accession Numbers of any nucleic acid sequences, protein sequences or atomic coordinates cited in the manuscript must be provided and include the corresponding database name.

- This information is provided on page 15

Authors Contributions:

Your 'Authors Contributions' section must detail the individual contribution for each individual author listed on your manuscript.

- This information is provided on page 15